Author's response to reviews

Title: Physical activity recommendations for health: what should Europe do?

Authors:

Pekka Oja (pekka.oja@uta.fi)
Fiona C Bull (F.C.L.Bull@lboro.ac.uk)
Mikael Fogelholm (Mikael.Fogelholm@aka.fi)
Brian W Martin (Brian.Martin@physicalactivityandhealth.ch)

Version: 3 Date: 23 December 2009

Author's response to reviews:

We have made the following changes:
1. The earlier headings "Current policy context", "Recent developments in physical activity recommendations", and "Is there a need for new or updated European physical activity recommendations?" have been deleted and the respective sections have been placed under the new heading "Discussion". Earlier headings have been incorporated into the text as lead sentences.
2. Reference 6 has been corrected.
3. Table 1 has been reformatted according to standard Word table form "Traditional 1". We hope it is acceptable now.
4. No typographical arrors to be changed have been found.